MOSS Meeting Minutes

June 10, 2003
ELEMENT ATTENDANCE:

ESDIS SSDO, ESDIS SOO, EDOS, EMOS, ASDC DAAC, TES SIPS, GES DAAC, MLS SIPS, HIRDLS SIPS, OMI SIPS, Averstar, MRTT

NEXT TELECON: June 24, 2003 at 2pm EST in Bldg 32 room S241

DIAL-IN NUMBER:

USA Toll Free Number: 888-395-8106

PASSCODE: 92854

MOSS 2 Preparation Schedule:
–May 19-June 6: MOSS 2 data delivery and preparation

–June 9-13: Write tapes for Polar Ground Stations

–June 16-17: Ship tapes to Polar Ground Stations

–July 1-2: (TBD) Abbreviated Dry run with Ground Stations

–New: July 8: MOSS 2 TRR
–July 14-18: Formal MOSS 2 test

MEETING HIGHLIGHTS:
1) Current MOSS 2 Preparation Schedule
See schedule above.

2) Mission Test #1 (MT1) – schedule indicates same week as MOSS2. If schedule holds, MOSS2 will need to be rescheduled. Karen will send out an email with some proposed dates for MOSS2, once the MT1 schedule firms up a little more. The dry runs for MT1 are occurring in June, with the last one being June 24th. EDOS mentioned that they do not participate in the MT1 dry runs; therefore none of the DAACs would participate.

3) GN contact plans for MOSS2: Day 1 only and Day 2 & 3 to be played locally by EDOS.

4) MOSS 2 Test Data Status:

a. Check days needed by SIPS; DOY 105-109?

Karen asked the teams if ephemeris for year 2003 DOY 105-109 noon-noon would be enough for their processing. All of the teams replied yes. One or two hours pass the end of their processing day would be sufficient. 

Eric reported that all of the test data have been received and processed by the test data group except for OMI. It was decided that, for OMI, SCIF2 data (including EDSs) would be timeshifted to be used for MOSS2. Action Item (AI): Jerry is to obtain the timelines and command sequences for SCIF2 and provide to Ted of OMI SIPS. After the telecon: Jerry provided the SCIF2 information to Ted. Eric also reported that the GBAD missing 8th package problem was fixed but there would be about a 3-hour gap in the data starting at the beginning of DOY 107.

5) SCIF3/CPT1 Data Evaluation and Processing Status:

a. HIRDLS –CPT1?

b. TES –SCIF3 eval/processing complete – output validation

c. OMI –CPT1 eval/processing complete? SCIF3 status

HIRDLS reported that they processed only a small portion of SCIF3 and CPT1 data and plans to process more of the data. TES validation of CPT1 output is complete. OMI completed CPT1 processing and reported checksum errors in SCIF3 processing. Known OMI hardware errors have been reported for SCIF3.

6) CPT2 Data delivery to DAACs -> SIPS

CPT2 data delivered today (6/10/03) to the DAACs -> SIPS. EDOS reported that all the RBDs were placed on the HIRDLS and OMI NRTs and HIRDLS was delivered via secure ftp. Nancy mentioned that the playbacks were for events 1, 3, and 4. Event 2 was corrupted and EDOS did not process it. TES initially wanted to obtain the event 2 data but after detail discussions of the problems with the data, TES decided not to request event 2 data.

GES DAAC reported that they were experiencing problems with subscriptions for MLS SCF and user services were looking into it. A lot of subscription related activities have been going on this week. ASDC DAAC reported that things appear to be normal for them and data were being distributed to TES.

OMI requested that ESDIS test data group generate and provide them with some L0 data (APIDs 1834, 1838, and 1840) with errors induced. The errors can be randomly generated and of all types. This activity would occur outside of MOSS activities.

7) QA MUT testing plans

It was decided that QA MUT would be tested in MOSS3. Steve mentioned that TES would require ESDT changes, as stated in an email from Steve to TES. Doug asked Steve to re-send the email to him.

8) Requirement changes

a. Removing references to auxiliary data flow

b. Ancillary changed to subscriptions

Karen briefly mentioned the recent changes to be applied to the requirements database.  For item 8b above, minor changes in requirement texts would occur.

9) New Toolkit 5.2.9 Release

Karen mentioned that the new Toolkit version 5.2.9 were released last week and wanted to make sure that everyone was aware of it. The Toolkit website also contains the release information.

Element Status:

EDOS

· Secure FTP implementation

· RBD path service – VCID 6

EDOS reported that the CPT2 RBDs delivered to HIRDLS NRTP were via secure ftp. Ron reported that there was a glitch in the first event, but subsequent events worked fine. They re-tried event 1 and it went through fine. HIRDLS was asked to look into it at their end to see if they saw any problems. Ron mentioned that the database group would be installing the changes for RBD path service for VCID 6 within the next two weeks. This should be ready by the first of July.

Ken of HIRDLS confirmed that APID 1630 would be archived at the GES DAAC. RBDs for VCIDs 6, 11, 3 (GBAD) will be distributed to both Boulder and Oxford.

EMOS/FOT

· History data plans – when ready to test?

EMOS reported that they are ready to test history data archive and retrieval. They are waiting for GES DAAC to be in OPS mode. Ed indicated that he would inform EMOS when GES DAAC is in OPS mode for Aura. Ed anticipates no sooner than July 1st.

FDS

· H/W status

· Need to schedule interface testing with GES DAAC (ICT3)

· Need expected delivery date for MOSS 2 products to GES DAAC

FDS was not present. After the telecon, Karen communicated with FDS via email.

ASDC DAAC

· Network issues?

Steve mentioned that he was waiting to hear from Jeff Smith on the current network changes status. The current plan was to complete the work by last week in June. Steve also stated that after the receipt of the CPT2 data, he would move everything over to haystack and would then use haystack for MOSS2. 

TES SIPS

· MTMGW

· MOSS1 testing ICD changes to Tom Goff/Stan Scott

· Req. # 1877 (produce L2) -- want to defer to MOSS3?

TES reported that they would start looking into MTMGW next week; the ICD changes are about ready to be sent to Tom Goff/Stan Scott, and they would like to defer reqs. #1876 and 1877 to MOSS3.  AI: TES to send their request for requirements changes to Karen via an email.
GES DAAC

· ESDTs Promotion to OPS Status:

· Receipt of .met files

· L0 ESDTs migration/promotion

· LeapSecT and UTCPoleT files going to SIPS?

James reported that he has received some .met files from every group. He believes that he has what he needs for the certification. If anything else is needed, James will contact the group independently. Ed stated that James was given the okay to write the CCR to promote the L0 ESDTs to OPS. Plans are to have it completed by July 14th. The ESDT promotion of the higher level products would follow the L0 promotion. If this plan holds, for MOSS2, the front end of the system would be in OPS mode and the back end would be in TS2 mode.

Ed mentioned that the SIPS should be receiving LeapSecT and UTCPoleT files. All of the teams confirmed receipt. However, MLS reported a discrepancy that the product version of the UTCPoleT and LeapSecT files received from the GES DAAC has been changed from 001 to 101, without notification to the MLS SIPS. AI: Ed to look into the UTCPoleT and LeapSecT files product version problem and get back to MLS.
Ed mentioned that he plans re-ingest the CPT2 data to kick off all the subscriptions to assist in verifying the subscription related changes.

MLS SIPS

· QA ready for MOSS2?

· Informal testing status with GES DAAC

· ML2IWC, ML2RHI -- want to defer to MOSS3?

Dave mentioned that he would need a sample metadata file for QA and afterwards they would be ready to test QA. Karen agreed to get a sample from the Science Office. After the telecon, Karen check with the Science Office and found out that for the system QA ESDT, a metadata file is not required. She sent out an email to the Aura SIPS-DAAC list on June 11th with this information and a sample metadata file was attached in the event that someone wants to generate one.

Audrey reported that their informal testing has started but is not complete. A system crash resulted in some lost of data, configuration problems encountered but fixed and CPT2 data transfer from the DAAC interrupted the testing and took priority. Audrey commended Latha from the GES DAAC for her excellent support. Audrey will continue to coordinate with the DAAC for completion of their informal tests. Audrey also confirmed that requirements for products ML2IWC and ML2RHI should be deferred to MOSS3. MLS requested the configuration information for MTMGW in OPS mode to be provided to them. AI: Ed to provide the OPS mode MTMGW information to Audrey.
HIRDLS SIPS/NRTP

· Informal testing status with GES DAAC of new products

Vince mentioned that he wants to setup time with the GES DAAC for some engineering testing of the new products (L1CAL, L2s, QA, etc.) They were going to try for Thursday (June 12th). Vince also reported that he got an error last week when trying to reorder data through the MTMGW. Ed mentioned that they may have been down during that time and asked Vince to try it again.

OMI SIPS

· Informal testing status with GES DAAC of new products

· MTMGW

· Subscription requests status all new MOSS2 products (DAO, NCEP, MODIS, etc.)

Marghi stated that OMI SIPS has not schedule any informal testing yet, including MTMGW, but would like to this week. She will contact the GES DAAC off-line to set up a time. OMI SIPS is finalizing their subscription requests. 

MRTT

Susan mentioned that MRTT would need the tapes number, from ESDIS, for the ground station once the data tapes are created.

ACTION ITEMS: Documented in the Aura Action Item Tool at http://jupiter02.gsfc.nasa.gov/auratesting/ . They will be highlighted in the minutes using the convention -- Action Item (AI): the text; but the status will be tracked via the tool. Please remember to enter your status in the tool.

